Author’s response to reviews

Title: External validation of molecular subtype classifications of colorectal cancer based on microsatellite instability, CIMP, BRAF and KRAS

Authors:

Elizabeth Alwers (elizabeth.alwers@dkfz.de)
Hendrik Bläker (Hendrik.Blaeker@charite.de)
Viola Walter (viowa86@gmail.com)
Lina Jansen (l.jansen@dkfz.de)
Matthias Kloor (Matthias.Kloor@med.uni-heidelberg.de)
Alexander Arnold (alexander.arnold@charite.de)
Julia Sieber-Frank (j.sieber-frank@t-online.de)
Esther Herpel (esther.herpel@med.uni-heidelberg.de)
Katrin Tagscherer (katrin.tagscherer@unimedizin-mainz.de)
Wilfried Roth (wilfried.roth@unimedizin-mainz.de)
Jenny Chang-Claude (j.chang-claude@dkfz-heidelberg.de)
Hermann Brenner (h.brenner@dkfz-heidelberg.de)
Michael Hoffmeister (m.hoffmeister@dkfz-heidelberg.de)

Version: 2 Date: 12 Jun 2019

Author’s response to reviews:

Editor Comments:

1 - In the Funding section of the Declarations please indicate the role of the funding body in the design of the study and collection, analysis, and interpretation of data and in writing the manuscript. If no specific funding was received for this study, please clearly indicate this in the Funding section.
We have indicated that the funding organizations had no role in the collection, analysis or interpretation of data, nor in any stage of the manuscript preparation (page 12).

2 - Please provide specific Authors' contributions for all of the authors; one author is missing from the statement; J S-F.

We have corrected this oversight and added the missing author to the appropriate contributions (page 12).

3 - At this stage, please upload your proofread manuscript as a single, final, clean version that does not contain any tracked changes, comments, highlights, strikethrough or text in different colours. All relevant tables and figures should also be clean versions. Figures (and additional files) should remain uploaded as separate files. Should you wish to respond to these revision requests, please include the information in the designated input box only.

We have removed all tracked changes and highlights from the manuscript. All tables and figures are submitted as clean versions.

4. Please include a statement in the competing interests to the effect that Author MK is an Editorial Board member for the journal.

We have added the corresponding statement.